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Seasonal Efect  on Power Generation in Colombia  
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•  Dry	
  condi*on	
  	
   •  Rainy	
  	
  condi*on	
  	
  

•  Some	
  Bagasse	
  is	
  being	
  	
  changed	
  for	
  coal	
  with	
  paper	
  mills.	
  

•  The	
  electricity	
  energy	
  prices	
  for	
  2010	
  were	
  between	
  117	
  $/kWh	
  
-­‐	
  126	
  $/kWh	
  (0.066	
  US$/kWh).	
  

•  Regula*ons	
  limit	
  Cogenera*on	
  to	
  max.	
  20	
  Mw. 	
  	
  	
  

Other conditions 



Cogeneration: Current situation 
of the Colombian sugar industry 

  9 of the 13 Colombian sugar mills are considered 
cogenerators, of these only 3 have technology to 
sell more of 15 Mwh to the grid. 

 
  6 can sell up to  1.5 Mwh. 

  Several cogeneration projects are under 
development to allow the Colombian sugar 
industry to sell more than 101 MWh, about 727 
GWh / year  



 Base Line: Mass and energy balances under current 

conditions. 

 Reduction of steam consumption through energy 

integration. 

 Defining  electrification level. 

 Cogeneration potential. 

Basic Engineering 
Methodology  



Energy Audits 

  Gross  and detailed mass 
and energy balances are 

performed as an Energy 

Audit  toghether with 

calculations using 

CENIPROF.  

  

AUDITORÍA  
ENERGÉTICA 

Generación  
Vapor Vivo 

Distribución  
Vapor Vivo 

Molienda 

Planta 
eléctrica 

Elaboración 

- Eficiencia Caldera 
 
- Consumo de Bagazo 
 
- Factores que afectan 
el rendimiento de las 
calderas 
 
Seguimientos a los 
consumos de vapor en: 
 
- Molinos 
- Turbogeneradores. 
- Destilería. 
- reductoras de vapor 
 
 
 
- Consumo de energía 
Mecánica. 
 
- Eficiencia térmica y 
Steam Rate de 
Turbinas. 
 
- Eficiencia de 
extracción  
 
 
 
 
- Seguimiento a la 
generación y 
distribución de energía 
eléctrica  
 
 
 Consumos de vapor en 
elaboración que 
incluyen: 
 
- Calentadores 
- Evaporadores 
- Tachos 
 
 

Área de Proceso Actividades 



CENIPROF  



Mass and energy balances. 

  Sankey Diagram, showing thermal energy distribution from fuels  to low 

pressure steam. 



Input and output  variables for  boiler efficiency assessment 



PCS Fuel 
100- Losses 

P1      P2      P3       P4       P5      P6 

Direct and Indirect Methods for fuel requirements 
determination 



Boiler efficiency assessment. 

59,5%
62,7%

56,9%
54,5%

58,3%
62,2%

80,0%

59,7%
62,2%

58,4%

65,5%

58,7%

49,8%

79,0%

56,2%
59,2%

72,1%
76,1%

73,1%

68,0%

62,4%63,8%

0,0%

10,0%

20,0%

30,0%

40,0%

50,0%

60,0%

70,0%

80,0%

90,0%

R1
RP 1

RP 3
CR 1

PC 2
MN 1

MN 3
CS 1

CB 1
CA 2

CA 4

bo
ile

r e
ffi

ci
en

cy
 (%

)

59
,5

%
62

,7
%

56
,9

%
54

,5
%

58
,3

%
62

,2
%

80
,0

%

59
,7

%
62

,2
%

58
,4

%

65
,5

%

58
,7

%

49
,8

%

79
,0

%

56
,2

%
59

,2
%

72
,1

%
76

,1
%

73
,1

%

68
,0

%

62
,4

%
63

,8
%

0,
0%

10
,0

%

20
,0

%

30
,0

%

40
,0

%

50
,0

%

60
,0

%

70
,0

%

80
,0

%

90
,0

%

R1

RP 1

RP 3

CR 1

PC 2

MN 1

MN 3

CS 1

CB 1

CA 2

CA 4

boiler efficiency (%)

59
,5

%
62

,7
%

56
,9

%
54

,5
%

58
,3

%
62

,2
%

80
,0

%

59
,7

%
62

,2
%

58
,4

%

65
,5

%

58
,7

%

49
,8

%

79
,0

%

56
,2

%
59

,2
%

72
,1

%
76

,1
%

73
,1

%

68
,0

%

62
,4

%
63

,8
%

0,
0%

10
,0

%

20
,0

%

30
,0

%

40
,0

%

50
,0

%

60
,0

%

70
,0

%

80
,0

%

90
,0

%

R1

RP 1

RP 3

CR 1

PC 2

MN 1

MN 3

CS 1

CB 1

CA 2

CA 4

boiler efficiency (%)

Boilers using coal and bagasse. Different ratios. 



Some Results 

 

  With the results of the energy audit a Base Line is established. 

  Three factors were found as critical for energy performance of 

the mill…. 

  Bagasse  HV  (moisture and ash ) affected by weather and 

operation. 

  Lost time. 

  Automatic Control strategies. 



Cogeneration 
 



Conditions studied for a mill 

  Case 1: Current conditions. 

  Case 2: Energy Integration for current conditions- 

  Case 3: Energy Integration  for new crushing rate and Imbibition 

280%  fibre .   

  Case 4: Energy Integration  for new crushing rate and  350%  

fibre Imbibition . Includes cogeneration project.  



CASE 1-2:  Current conditions-Some energy integration 

BALANCE DE MASA Y ENERGÍA CONDICIONES 2009 INGENIO MANUELITA

Consumos de Baja Presión

Tandem I Tandem II Tipo de vapor
Calentador Pre-primario 158.1 218.4 Condensado
Calentador Primario 5.9 11.7 Gases III
Calentador Secundario 12.0 23.6 Gases II
Calentador JC Primario Gases I
Calentador JC secundario Escape
Calentador de Licor Gases III

Tachos de Crudo Gases III
Tachos Refinería Gases II

Secadoras Gases I

Evaporador 1 Escape
Destilería Escape

Consumos de Alta Presión

Turbogenerador Vivo
Destilería Vivo
Consumo especifico Vapor % Caña
Consumo especifico lb vapor/ton caña

Condiciones Vapor

Presión vapor vivo Psig
Temperatura Vapor vivo °C
Presión Vapor escape Psig
Temperatura Vapor escape °C

Área de Evaporación

Evaporador 1 m2
Evaporador 2 m2
Evaporador 3 m2
Evaporador 4 m2
Evaporador 5 m2

1067

33.7

3.9

170.1
26.7

Modelo Ceniprof

207.0
4.1

48.4%

5.3
11.6
2.0

50.0

308

5403
7306
7442
194

950
480
24
128

Steam consumption 



CASE 3: New crushing rate , Energetic integration. 

 
  500 TCH, fiber 16% cane. 
  Imbibition 280% fiber. 

  All Electric drives. 
  Low pressure steam as Case 2.  

  New Boiler (950 psig – 480°C) 

 

Steam consumption- Refinery and ethanol plant 

BALANCE DE MASA Y ENERGÍA NUEVAS CONDICIONES INGENIO MANUELITA

Consumos de Baja Presión

Tandem I Tandem II Tipo de vapor
Calentador Pre-primario 169,6 253,2 ton/h Condensado
Calentador Primario 6,4 14,8 ton/h Gases III
Calentador Secundario 12,6 29,5 ton/h Gases II
Calentador JC Primario ton/h Gases I
Calentador JC secundario ton/h Escape
Calentador de Licor ton/h Gases III

Tachos de Crudo ton/h Gases III
Tachos Refinería ton/h Gases II

Secadoras ton/h Gases I

Evaporador 1 ton/h Escape
Destilería ton/h Escape

Consumos de Alta Presión

Turbogenerador ton/h Vivo
Destilería ton/h Vivo
Consumo especifico ton/h Vapor % Caña
Consumo especifico ton/h lb vapor/ton caña

Condiciones Vapor

Presión vapor vivo Psig
Temperatura Vapor vivo °C
Presión Vapor escape Psig
Temperatura Vapor escape °C

Área de Evaporación

Evaporador 1 m2
Evaporador 2 m3
Evaporador 3 m4
Evaporador 4 m5
Evaporador 5 m6821

6053
7876
7265
678

950
480
24
128

Modelo Ceniprof

237,4
4,8

48,4%

6,3
13,8
2,3

51,6

1067

33,7

3,9

194,2
31,0

Required area 

BALANCE DE MASA Y ENERGÍA NUEVAS CONDICIONES INGENIO MANUELITA

Consumos de Baja Presión

Tandem I Tandem II Tipo de vapor
Calentador Pre-primario 169,6 253,2 ton/h Condensado
Calentador Primario 6,4 14,8 ton/h Gases III
Calentador Secundario 12,6 29,5 ton/h Gases II
Calentador JC Primario ton/h Gases I
Calentador JC secundario ton/h Escape
Calentador de Licor ton/h Gases III

Tachos de Crudo ton/h Gases III
Tachos Refinería ton/h Gases II

Secadoras ton/h Gases I

Evaporador 1 ton/h Escape
Destilería ton/h Escape

Consumos de Alta Presión

Turbogenerador ton/h Vivo
Destilería ton/h Vivo
Consumo especifico ton/h Vapor % Caña
Consumo especifico ton/h lb vapor/ton caña

Condiciones Vapor

Presión vapor vivo Psig
Temperatura Vapor vivo °C
Presión Vapor escape Psig
Temperatura Vapor escape °C

Área de Evaporación

Evaporador 1 m2
Evaporador 2 m3
Evaporador 3 m4
Evaporador 4 m5
Evaporador 5 m6821

6053
7876
7265
678

950
480
24
128

Modelo Ceniprof

237,4
4,8

48,4%

6,3
13,8
2,3

51,6

1067

33,7

3,9

194,2
31,0

Bagasse surplus 
 

Generación Vapor

Caldera Nueva ton/h
Caldera 7 ton/h
Caldera 5 ton/h

Vapor % Caña %

Consumo Bagazo

Modelo Ceniprof
Caldera Nueva 125,0 ton/h

Caldera 7 0 ton/h
Caldera 5 0 ton/h

Bagazo % Caña 0 %
Bagazo producido 161 ton/h
Bagazo excedente 36 ton/h
Bagazo Producido 1199574 ton/año

Bagazo Consumido 960328 ton/año
Excedentes de bzo 239247 ton/año

1148

Modelo Ceniprof
242,2
0
0

32, 

Estimated 1.94 lb vapor/lb bagasse 



CASE 4: New crushing rate,Energy integration and 
Cogeneration- 

 
  500 TCH, fibre 16.0% 
  Imbibition 350% 

  Total electric drives 

  Energy integration Case 2 

  Steam Boiler 950 psig – 480ºC 

Steam consumption 

Evaporators area required 

Electricity 
 

Consumos de Baja Presión

Tandem I Tandem II Tipo de vapor
Calentador Pre-primario 0.0 326.1 ton/h Condensado
Calentador Primario 0.0 35.1 ton/h Gases III
Calentador Secundario 0.0 34.0 ton/h Gases II
Calentador JC Primario ton/h Gases I
Calentador JC secundario ton/h Escape
Calentador de Licor ton/h Gases III

Tachos de Crudo ton/h Gases III
Tachos Refinería ton/h Gases II

Secadoras ton/h Gases I

Evaporador 1 ton/h Escape
Destilería ton/h Escape

Consumos de Alta Presión

Turbogenerador ton/h Vivo
Destilería ton/h Vivo
Consumo especifico ton/h Vapor % Caña
Consumo especifico ton/h lb vapor/ton caña

Condiciones Vapor

Presión vapor vivo Psig
Temperatura Vapor vivo °C
Presión Vapor escape Psig
Temperatura Vapor escape °C

Área de Evaporación

Evaporador 1 m2
Evaporador 2 m3
Evaporador 3 m4
Evaporador 4 m5
Evaporador 5 m6

* Temperatura de Jugo salida de calentamiento primario 80 °C

1136

33.7

3.9

208.1
31.3

Modelo Ceniprof

252.9
4.8

51.5%

7.0
15.2
2.3

51.8

950
480
24
128

1136

6240
7869
8156
951

Consumos de Baja Presión

Tandem I Tandem II Tipo de vapor
Calentador Pre-primario 0.0 326.1 ton/h Condensado
Calentador Primario 0.0 35.1 ton/h Gases III
Calentador Secundario 0.0 34.0 ton/h Gases II
Calentador JC Primario ton/h Gases I
Calentador JC secundario ton/h Escape
Calentador de Licor ton/h Gases III

Tachos de Crudo ton/h Gases III
Tachos Refinería ton/h Gases II

Secadoras ton/h Gases I

Evaporador 1 ton/h Escape
Destilería ton/h Escape

Consumos de Alta Presión

Turbogenerador ton/h Vivo
Destilería ton/h Vivo
Consumo especifico ton/h Vapor % Caña
Consumo especifico ton/h lb vapor/ton caña

Condiciones Vapor

Presión vapor vivo Psig
Temperatura Vapor vivo °C
Presión Vapor escape Psig
Temperatura Vapor escape °C

Área de Evaporación

Evaporador 1 m2
Evaporador 2 m3
Evaporador 3 m4
Evaporador 4 m5
Evaporador 5 m6

* Temperatura de Jugo salida de calentamiento primario 80 °C

1136

33.7

3.9

208.1
31.3

Modelo Ceniprof

252.9
4.8

51.5%

7.0
15.2
2.3

51.8

950
480
24
128

1136

6240
7869
8156
951

BALANCE DE MASA Y ENERGÍA CONDICIONES FUTURAS INGENIO MANUELITA

Consumos de Baja Presión

Tandem I Tandem II Tipo de vapor
Calentador Pre-primario 0,0 326,1 ton/h Condensado
Calentador Primario 0,0 35,1 ton/h Gases III
Calentador Secundario 0,0 34,0 ton/h Gases II
Calentador JC Primario ton/h Gases I
Calentador JC secundario ton/h Escape
Calentador de Licor ton/h Gases III

Tachos de Crudo ton/h Gases III
Tachos Refinería ton/h Gases II

Secadoras ton/h Gases I

Evaporador 1 ton/h Escape
Destilería ton/h Escape

Consumos de Alta Presión

Turbogenerador Contra ton/h Vivo
Turbogenerador Condens Vivo
Destilería ton/h Vivo
Consumo especifico ton/h Vapor % Caña
Consumo especifico ton/h lb vapor/ton caña

Condiciones Vapor

Presión vapor vivo Psig
Temperatura Vapor vivo °C
Presión Vapor escape Psig
Temperatura Vapor escape °C

Área de Evaporación

Evaporador 1 m2
Evaporador 2 m3
Evaporador 3 m4
Evaporador 4 m5
Evaporador 5 m6

* Temperatura de Jugo salida de calentamiento primario 80 °C

Generación de energía Eléctrica

Turbogenerador Contra kWh
Turbogenerador Condens kWh
Total kWh
Steam rate Turbo Contrapresión lb vapor/kWh13,0

43016

1136

128,16

1136

43016

951

6240
7869
8156

950
480
24

0

Modelo Ceniprof

252,9

4,8

7,0
15,2
2,3

51,8

0,0

51,5%

33,7

3,9

208,1
31,3

•  Internal electricity consumption 20500 kWh. 

•  Available to the grid22516 kWh. 
•  Bagasse surplus : 29.5 ton/h. 



CASE 4: SENSITIVITY 

•  Bagasse surplus to  fiber % cane 
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•  Bagasse surplus to fuel quality 
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•  Specific steam consumption vs. 

Lost time.  



Summary 

  It is possible to sell to the grid > 20000 kWh  with only bagasse. 

  Effects of weather- cane quality and  lost time  were considered. 





Boilers 



Power Plant 



Electric Sub/Station 



Milling train drives 



Generated Power  
01/2005  to 03/2011 



Power  to the grid 
01/2005  to 03/2011 



Balance of Electricity-Bagasse-
Coal (Col$x1000)  



Kyoto Protocol Analysis 



Total Cost for a Cogeneration 
Project 



THANKS…. 


